UNIX Course Account Form for 112b

Department of Computer Science, Smith College, Northampton, Massachusetts 01063

Here is your Computer Science UNIX account. Your account name is: 112b

Your password is: obQ’t61A

Read the material on ‘‘System Practices,’’ sign the agreement below, tear it off along the dotted line, and return it to your instructor.

System Practices

Smith instructional computer systems are meant to be used for any and every educational purpose. There are no fixed limits on or charges for login time, use of the CPU (processor), or file storage space. Since some resources can be used up—printer paper, or access to terminals—we ask that you use these only as necessary. Help keep the systems unregulated—don’t waste or monopolize resources. When file storage space runs out, pitch in to help make space available. Similarly, use your good judgement to minimize use of CPU-intensive programs when the system is heavily loaded. Use the system as much as you need to, but let other people work too. Beowulf, grendel, and the other Computer Science machines are run as an open system. The UNIX philosophy is to leave as much of the computer open for inspection as possible. One of the best ways to learn what’s available is to explore. Secrecy is not the standard on UNIX. However, while system files are public, other people’s files are not. Respect the privacy of other users. This account is for your use only; it may not be given away or shared.

Getting Started

After you first log on, you might want to change your given, random password; and you might want to update your ‘‘finger’’information. Be forewarned that passwd will check for and reject easily cracked passwords. To change your password, type: passwd To change your ‘‘finger’’ information, type: chfn To logout, type: logout A few other Unix commands: To check your finger information, type: finger 112b To see manual page for command, type: man command To list commands related to keyword, type: man -k keyword To list a text file, type: more To read mail, type: pine To read news, type: tin Other commands (use man to learn more): ls, chmod, finger, w, mkdir, cd, mv, emacs, vi ........................................................................................................................................................................ Return this portion to your instructor. Account name: 112b

In accepting this account, I agree to follow the system practices outlined above.

Name (please print):

Campus Box: Phone Number:

Signature:

UNIX Course Account Form for 112b

Department of Computer Science, Smith College, Northampton, Massachusetts 01063

Here is your Computer Science UNIX account. Your account name is: 112b-t1

Your password is: r*eYV81z

Read the material on ‘‘System Practices,’’ sign the agreement below, tear it off along the dotted line, and return it to your instructor.

System Practices

Smith instructional computer systems are meant to be used for any and every educational purpose. There are no fixed limits on or charges for login time, use of the CPU (processor), or file storage space. Since some resources can be used up—printer paper, or access to terminals—we ask that you use these only as necessary. Help keep the systems unregulated—don’t waste or monopolize resources. When file storage space runs out, pitch in to help make space available. Similarly, use your good judgement to minimize use of CPU-intensive programs when the system is heavily loaded. Use the system as much as you need to, but let other people work too. Beowulf, grendel, and the other Computer Science machines are run as an open system. The UNIX philosophy is to leave as much of the computer open for inspection as possible. One of the best ways to learn what’s available is to explore. Secrecy is not the standard on UNIX. However, while system files are public, other people’s files are not. Respect the privacy of other users. This account is for your use only; it may not be given away or shared.

Getting Started

After you first log on, you might want to change your given, random password; and you might want to update your ‘‘finger’’information. Be forewarned that passwd will check for and reject easily cracked passwords. To change your password, type: passwd To change your ‘‘finger’’ information, type: chfn To logout, type: logout A few other Unix commands: To check your finger information, type: finger 112b-t1 To see manual page for command, type: man command To list commands related to keyword, type: man -k keyword To list a text file, type: more To read mail, type: pine To read news, type: tin Other commands (use man to learn more): ls, chmod, finger, w, mkdir, cd, mv, emacs, vi ........................................................................................................................................................................ Return this portion to your instructor. Account name: 112b-t1

In accepting this account, I agree to follow the system practices outlined above.

Name (please print):

Campus Box: Phone Number:

UNIX Course Account Form for 112b

Department of Computer Science, Smith College, Northampton, Massachusetts 01063

Here is your Computer Science UNIX account. Your account name is: 112b-t2

Your password is: qqT02$bS

Read the material on ‘‘System Practices,’’ sign the agreement below, tear it off along the dotted line, and return it to your instructor.

System Practices

Smith instructional computer systems are meant to be used for any and every educational purpose. There are no fixed limits on or charges for login time, use of the CPU (processor), or file storage space. Since some resources can be used up—printer paper, or access to terminals—we ask that you use these only as necessary. Help keep the systems unregulated—don’t waste or monopolize resources. When file storage space runs out, pitch in to help make space available. Similarly, use your good judgement to minimize use of CPU-intensive programs when the system is heavily loaded. Use the system as much as you need to, but let other people work too. Beowulf, grendel, and the other Computer Science machines are run as an open system. The UNIX philosophy is to leave as much of the computer open for inspection as possible. One of the best ways to learn what’s available is to explore. Secrecy is not the standard on UNIX. However, while system files are public, other people’s files are not. Respect the privacy of other users. This account is for your use only; it may not be given away or shared.

Getting Started

After you first log on, you might want to change your given, random password; and you might want to update your ‘‘finger’’information. Be forewarned that passwd will check for and reject easily cracked passwords. To change your password, type: passwd To change your ‘‘finger’’ information, type: chfn To logout, type: logout A few other Unix commands: To check your finger information, type: finger 112b-t2 To see manual page for command, type: man command To list commands related to keyword, type: man -k keyword To list a text file, type: more To read mail, type: pine To read news, type: tin Other commands (use man to learn more): ls, chmod, finger, w, mkdir, cd, mv, emacs, vi ........................................................................................................................................................................ Return this portion to your instructor. Account name: 112b-t2

In accepting this account, I agree to follow the system practices outlined above.

Name (please print):

Campus Box: Phone Number:

UNIX Course Account Form for 112b

Department of Computer Science, Smith College, Northampton, Massachusetts 01063

Here is your Computer Science UNIX account. Your account name is: 112b-aa

Your password is: 6.d8OrgM

Read the material on ‘‘System Practices,’’ sign the agreement below, tear it off along the dotted line, and return it to your instructor.

System Practices

Smith instructional computer systems are meant to be used for any and every educational purpose. There are no fixed limits on or charges for login time, use of the CPU (processor), or file storage space. Since some resources can be used up—printer paper, or access to terminals—we ask that you use these only as necessary. Help keep the systems unregulated—don’t waste or monopolize resources. When file storage space runs out, pitch in to help make space available. Similarly, use your good judgement to minimize use of CPU-intensive programs when the system is heavily loaded. Use the system as much as you need to, but let other people work too. Beowulf, grendel, and the other Computer Science machines are run as an open system. The UNIX philosophy is to leave as much of the computer open for inspection as possible. One of the best ways to learn what’s available is to explore. Secrecy is not the standard on UNIX. However, while system files are public, other people’s files are not. Respect the privacy of other users. This account is for your use only; it may not be given away or shared.

Getting Started

After you first log on, you might want to change your given, random password; and you might want to update your ‘‘finger’’information. Be forewarned that passwd will check for and reject easily cracked passwords. To change your password, type: passwd To change your ‘‘finger’’ information, type: chfn To logout, type: logout A few other Unix commands: To check your finger information, type: finger 112b-aa To see manual page for command, type: man command To list commands related to keyword, type: man -k keyword To list a text file, type: more To read mail, type: pine To read news, type: tin Other commands (use man to learn more): ls, chmod, finger, w, mkdir, cd, mv, emacs, vi ........................................................................................................................................................................ Return this portion to your instructor. Account name: 112b-aa

In accepting this account, I agree to follow the system practices outlined above.

Name (please print):

Campus Box: Phone Number:

UNIX Course Account Form for 112b

Department of Computer Science, Smith College, Northampton, Massachusetts 01063

Here is your Computer Science UNIX account. Your account name is: 112b-ab

Your password is: 6}nZj2Av

Read the material on ‘‘System Practices,’’ sign the agreement below, tear it off along the dotted line, and return it to your instructor.

System Practices

Smith instructional computer systems are meant to be used for any and every educational purpose. There are no fixed limits on or charges for login time, use of the CPU (processor), or file storage space. Since some resources can be used up—printer paper, or access to terminals—we ask that you use these only as necessary. Help keep the systems unregulated—don’t waste or monopolize resources. When file storage space runs out, pitch in to help make space available. Similarly, use your good judgement to minimize use of CPU-intensive programs when the system is heavily loaded. Use the system as much as you need to, but let other people work too. Beowulf, grendel, and the other Computer Science machines are run as an open system. The UNIX philosophy is to leave as much of the computer open for inspection as possible. One of the best ways to learn what’s available is to explore. Secrecy is not the standard on UNIX. However, while system files are public, other people’s files are not. Respect the privacy of other users. This account is for your use only; it may not be given away or shared.

Getting Started

After you first log on, you might want to change your given, random password; and you might want to update your ‘‘finger’’information. Be forewarned that passwd will check for and reject easily cracked passwords. To change your password, type: passwd To change your ‘‘finger’’ information, type: chfn To logout, type: logout A few other Unix commands: To check your finger information, type: finger 112b-ab To see manual page for command, type: man command To list commands related to keyword, type: man -k keyword To list a text file, type: more To read mail, type: pine To read news, type: tin Other commands (use man to learn more): ls, chmod, finger, w, mkdir, cd, mv, emacs, vi ........................................................................................................................................................................ Return this portion to your instructor. Account name: 112b-ab

In accepting this account, I agree to follow the system practices outlined above.

Name (please print):

Campus Box: Phone Number:

UNIX Course Account Form for 112b

Department of Computer Science, Smith College, Northampton, Massachusetts 01063

Here is your Computer Science UNIX account. Your account name is: 112b-ac

Your password is: Rg08wLe*

Read the material on ‘‘System Practices,’’ sign the agreement below, tear it off along the dotted line, and return it to your instructor.

System Practices

Smith instructional computer systems are meant to be used for any and every educational purpose. There are no fixed limits on or charges for login time, use of the CPU (processor), or file storage space. Since some resources can be used up—printer paper, or access to terminals—we ask that you use these only as necessary. Help keep the systems unregulated—don’t waste or monopolize resources. When file storage space runs out, pitch in to help make space available. Similarly, use your good judgement to minimize use of CPU-intensive programs when the system is heavily loaded. Use the system as much as you need to, but let other people work too. Beowulf, grendel, and the other Computer Science machines are run as an open system. The UNIX philosophy is to leave as much of the computer open for inspection as possible. One of the best ways to learn what’s available is to explore. Secrecy is not the standard on UNIX. However, while system files are public, other people’s files are not. Respect the privacy of other users. This account is for your use only; it may not be given away or shared.

Getting Started

After you first log on, you might want to change your given, random password; and you might want to update your ‘‘finger’’information. Be forewarned that passwd will check for and reject easily cracked passwords. To change your password, type: passwd To change your ‘‘finger’’ information, type: chfn To logout, type: logout A few other Unix commands: To check your finger information, type: finger 112b-ac To see manual page for command, type: man command To list commands related to keyword, type: man -k keyword To list a text file, type: more To read mail, type: pine To read news, type: tin Other commands (use man to learn more): ls, chmod, finger, w, mkdir, cd, mv, emacs, vi ........................................................................................................................................................................ Return this portion to your instructor. Account name: 112b-ac

In accepting this account, I agree to follow the system practices outlined above.

Name (please print):

Campus Box: Phone Number:

UNIX Course Account Form for 112b

Department of Computer Science, Smith College, Northampton, Massachusetts 01063

Here is your Computer Science UNIX account. Your account name is: 112b-ad

Your password is: AI09hxi=

Read the material on ‘‘System Practices,’’ sign the agreement below, tear it off along the dotted line, and return it to your instructor.

System Practices

Smith instructional computer systems are meant to be used for any and every educational purpose. There are no fixed limits on or charges for login time, use of the CPU (processor), or file storage space. Since some resources can be used up—printer paper, or access to terminals—we ask that you use these only as necessary. Help keep the systems unregulated—don’t waste or monopolize resources. When file storage space runs out, pitch in to help make space available. Similarly, use your good judgement to minimize use of CPU-intensive programs when the system is heavily loaded. Use the system as much as you need to, but let other people work too. Beowulf, grendel, and the other Computer Science machines are run as an open system. The UNIX philosophy is to leave as much of the computer open for inspection as possible. One of the best ways to learn what’s available is to explore. Secrecy is not the standard on UNIX. However, while system files are public, other people’s files are not. Respect the privacy of other users. This account is for your use only; it may not be given away or shared.

Getting Started

After you first log on, you might want to change your given, random password; and you might want to update your ‘‘finger’’information. Be forewarned that passwd will check for and reject easily cracked passwords. To change your password, type: passwd To change your ‘‘finger’’ information, type: chfn To logout, type: logout A few other Unix commands: To check your finger information, type: finger 112b-ad To see manual page for command, type: man command To list commands related to keyword, type: man -k keyword To list a text file, type: more To read mail, type: pine To read news, type: tin Other commands (use man to learn more): ls, chmod, finger, w, mkdir, cd, mv, emacs, vi ........................................................................................................................................................................ Return this portion to your instructor. Account name: 112b-ad

In accepting this account, I agree to follow the system practices outlined above.

Name (please print):

Campus Box: Phone Number:

UNIX Course Account Form for 112b

Department of Computer Science, Smith College, Northampton, Massachusetts 01063

Here is your Computer Science UNIX account. Your account name is: 112b-ae

Your password is: vbr"GC88

Read the material on ‘‘System Practices,’’ sign the agreement below, tear it off along the dotted line, and return it to your instructor.

System Practices

Smith instructional computer systems are meant to be used for any and every educational purpose. There are no fixed limits on or charges for login time, use of the CPU (processor), or file storage space. Since some resources can be used up—printer paper, or access to terminals—we ask that you use these only as necessary. Help keep the systems unregulated—don’t waste or monopolize resources. When file storage space runs out, pitch in to help make space available. Similarly, use your good judgement to minimize use of CPU-intensive programs when the system is heavily loaded. Use the system as much as you need to, but let other people work too. Beowulf, grendel, and the other Computer Science machines are run as an open system. The UNIX philosophy is to leave as much of the computer open for inspection as possible. One of the best ways to learn what’s available is to explore. Secrecy is not the standard on UNIX. However, while system files are public, other people’s files are not. Respect the privacy of other users. This account is for your use only; it may not be given away or shared.

Getting Started

After you first log on, you might want to change your given, random password; and you might want to update your ‘‘finger’’information. Be forewarned that passwd will check for and reject easily cracked passwords. To change your password, type: passwd To change your ‘‘finger’’ information, type: chfn To logout, type: logout A few other Unix commands: To check your finger information, type: finger 112b-ae To see manual page for command, type: man command To list commands related to keyword, type: man -k keyword To list a text file, type: more To read mail, type: pine To read news, type: tin Other commands (use man to learn more): ls, chmod, finger, w, mkdir, cd, mv, emacs, vi ........................................................................................................................................................................ Return this portion to your instructor. Account name: 112b-ae

In accepting this account, I agree to follow the system practices outlined above.

Name (please print):

Campus Box: Phone Number:

UNIX Course Account Form for 112b

Department of Computer Science, Smith College, Northampton, Massachusetts 01063

Here is your Computer Science UNIX account. Your account name is: 112b-af

Your password is: w0EZmm7~

Read the material on ‘‘System Practices,’’ sign the agreement below, tear it off along the dotted line, and return it to your instructor.

System Practices

Smith instructional computer systems are meant to be used for any and every educational purpose. There are no fixed limits on or charges for login time, use of the CPU (processor), or file storage space. Since some resources can be used up—printer paper, or access to terminals—we ask that you use these only as necessary. Help keep the systems unregulated—don’t waste or monopolize resources. When file storage space runs out, pitch in to help make space available. Similarly, use your good judgement to minimize use of CPU-intensive programs when the system is heavily loaded. Use the system as much as you need to, but let other people work too. Beowulf, grendel, and the other Computer Science machines are run as an open system. The UNIX philosophy is to leave as much of the computer open for inspection as possible. One of the best ways to learn what’s available is to explore. Secrecy is not the standard on UNIX. However, while system files are public, other people’s files are not. Respect the privacy of other users. This account is for your use only; it may not be given away or shared.

Getting Started

After you first log on, you might want to change your given, random password; and you might want to update your ‘‘finger’’information. Be forewarned that passwd will check for and reject easily cracked passwords. To change your password, type: passwd To change your ‘‘finger’’ information, type: chfn To logout, type: logout A few other Unix commands: To check your finger information, type: finger 112b-af To see manual page for command, type: man command To list commands related to keyword, type: man -k keyword To list a text file, type: more To read mail, type: pine To read news, type: tin Other commands (use man to learn more): ls, chmod, finger, w, mkdir, cd, mv, emacs, vi ........................................................................................................................................................................ Return this portion to your instructor. Account name: 112b-af

In accepting this account, I agree to follow the system practices outlined above.

Name (please print):

Campus Box: Phone Number:

UNIX Course Account Form for 112b

Department of Computer Science, Smith College, Northampton, Massachusetts 01063

Here is your Computer Science UNIX account. Your account name is: 112b-ag

Your password is: 8Dp0Cob{

Read the material on ‘‘System Practices,’’ sign the agreement below, tear it off along the dotted line, and return it to your instructor.

System Practices

Smith instructional computer systems are meant to be used for any and every educational purpose. There are no fixed limits on or charges for login time, use of the CPU (processor), or file storage space. Since some resources can be used up—printer paper, or access to terminals—we ask that you use these only as necessary. Help keep the systems unregulated—don’t waste or monopolize resources. When file storage space runs out, pitch in to help make space available. Similarly, use your good judgement to minimize use of CPU-intensive programs when the system is heavily loaded. Use the system as much as you need to, but let other people work too. Beowulf, grendel, and the other Computer Science machines are run as an open system. The UNIX philosophy is to leave as much of the computer open for inspection as possible. One of the best ways to learn what’s available is to explore. Secrecy is not the standard on UNIX. However, while system files are public, other people’s files are not. Respect the privacy of other users. This account is for your use only; it may not be given away or shared.

Getting Started

After you first log on, you might want to change your given, random password; and you might want to update your ‘‘finger’’information. Be forewarned that passwd will check for and reject easily cracked passwords. To change your password, type: passwd To change your ‘‘finger’’ information, type: chfn To logout, type: logout A few other Unix commands: To check your finger information, type: finger 112b-ag To see manual page for command, type: man command To list commands related to keyword, type: man -k keyword To list a text file, type: more To read mail, type: pine To read news, type: tin Other commands (use man to learn more): ls, chmod, finger, w, mkdir, cd, mv, emacs, vi ........................................................................................................................................................................ Return this portion to your instructor. Account name: 112b-ag

In accepting this account, I agree to follow the system practices outlined above.

Name (please print):

Campus Box: Phone Number:

UNIX Course Account Form for 112b

Department of Computer Science, Smith College, Northampton, Massachusetts 01063

Here is your Computer Science UNIX account. Your account name is: 112b-ah

Your password is: N"5Tymq3

Read the material on ‘‘System Practices,’’ sign the agreement below, tear it off along the dotted line, and return it to your instructor.

System Practices

Smith instructional computer systems are meant to be used for any and every educational purpose. There are no fixed limits on or charges for login time, use of the CPU (processor), or file storage space. Since some resources can be used up—printer paper, or access to terminals—we ask that you use these only as necessary. Help keep the systems unregulated—don’t waste or monopolize resources. When file storage space runs out, pitch in to help make space available. Similarly, use your good judgement to minimize use of CPU-intensive programs when the system is heavily loaded. Use the system as much as you need to, but let other people work too. Beowulf, grendel, and the other Computer Science machines are run as an open system. The UNIX philosophy is to leave as much of the computer open for inspection as possible. One of the best ways to learn what’s available is to explore. Secrecy is not the standard on UNIX. However, while system files are public, other people’s files are not. Respect the privacy of other users. This account is for your use only; it may not be given away or shared.

Getting Started

After you first log on, you might want to change your given, random password; and you might want to update your ‘‘finger’’information. Be forewarned that passwd will check for and reject easily cracked passwords. To change your password, type: passwd To change your ‘‘finger’’ information, type: chfn To logout, type: logout A few other Unix commands: To check your finger information, type: finger 112b-ah To see manual page for command, type: man command To list commands related to keyword, type: man -k keyword To list a text file, type: more To read mail, type: pine To read news, type: tin Other commands (use man to learn more): ls, chmod, finger, w, mkdir, cd, mv, emacs, vi ........................................................................................................................................................................ Return this portion to your instructor. Account name: 112b-ah

In accepting this account, I agree to follow the system practices outlined above.

Name (please print):

Campus Box: Phone Number:

UNIX Course Account Form for 112b

Department of Computer Science, Smith College, Northampton, Massachusetts 01063

Here is your Computer Science UNIX account. Your account name is: 112b-ai

Your password is: hhM95%tW

Read the material on ‘‘System Practices,’’ sign the agreement below, tear it off along the dotted line, and return it to your instructor.

System Practices

Smith instructional computer systems are meant to be used for any and every educational purpose. There are no fixed limits on or charges for login time, use of the CPU (processor), or file storage space. Since some resources can be used up—printer paper, or access to terminals—we ask that you use these only as necessary. Help keep the systems unregulated—don’t waste or monopolize resources. When file storage space runs out, pitch in to help make space available. Similarly, use your good judgement to minimize use of CPU-intensive programs when the system is heavily loaded. Use the system as much as you need to, but let other people work too. Beowulf, grendel, and the other Computer Science machines are run as an open system. The UNIX philosophy is to leave as much of the computer open for inspection as possible. One of the best ways to learn what’s available is to explore. Secrecy is not the standard on UNIX. However, while system files are public, other people’s files are not. Respect the privacy of other users. This account is for your use only; it may not be given away or shared.

Getting Started

After you first log on, you might want to change your given, random password; and you might want to update your ‘‘finger’’information. Be forewarned that passwd will check for and reject easily cracked passwords. To change your password, type: passwd To change your ‘‘finger’’ information, type: chfn To logout, type: logout A few other Unix commands: To check your finger information, type: finger 112b-ai To see manual page for command, type: man command To list commands related to keyword, type: man -k keyword To list a text file, type: more To read mail, type: pine To read news, type: tin Other commands (use man to learn more): ls, chmod, finger, w, mkdir, cd, mv, emacs, vi ........................................................................................................................................................................ Return this portion to your instructor. Account name: 112b-ai

In accepting this account, I agree to follow the system practices outlined above.

Name (please print):

Campus Box: Phone Number:

UNIX Course Account Form for 112b

Department of Computer Science, Smith College, Northampton, Massachusetts 01063

Here is your Computer Science UNIX account. Your account name is: 112b-aj

Your password is: hsj-1GL7

Read the material on ‘‘System Practices,’’ sign the agreement below, tear it off along the dotted line, and return it to your instructor.

System Practices

Smith instructional computer systems are meant to be used for any and every educational purpose. There are no fixed limits on or charges for login time, use of the CPU (processor), or file storage space. Since some resources can be used up—printer paper, or access to terminals—we ask that you use these only as necessary. Help keep the systems unregulated—don’t waste or monopolize resources. When file storage space runs out, pitch in to help make space available. Similarly, use your good judgement to minimize use of CPU-intensive programs when the system is heavily loaded. Use the system as much as you need to, but let other people work too. Beowulf, grendel, and the other Computer Science machines are run as an open system. The UNIX philosophy is to leave as much of the computer open for inspection as possible. One of the best ways to learn what’s available is to explore. Secrecy is not the standard on UNIX. However, while system files are public, other people’s files are not. Respect the privacy of other users. This account is for your use only; it may not be given away or shared.

Getting Started

After you first log on, you might want to change your given, random password; and you might want to update your ‘‘finger’’information. Be forewarned that passwd will check for and reject easily cracked passwords. To change your password, type: passwd To change your ‘‘finger’’ information, type: chfn To logout, type: logout A few other Unix commands: To check your finger information, type: finger 112b-aj To see manual page for command, type: man command To list commands related to keyword, type: man -k keyword To list a text file, type: more To read mail, type: pine To read news, type: tin Other commands (use man to learn more): ls, chmod, finger, w, mkdir, cd, mv, emacs, vi ........................................................................................................................................................................ Return this portion to your instructor. Account name: 112b-aj

In accepting this account, I agree to follow the system practices outlined above.

Name (please print):

Campus Box: Phone Number:

UNIX Course Account Form for 112b

Department of Computer Science, Smith College, Northampton, Massachusetts 01063

Here is your Computer Science UNIX account. Your account name is: 112b-ak

Your password is: p6Iw0Ja!

Read the material on ‘‘System Practices,’’ sign the agreement below, tear it off along the dotted line, and return it to your instructor.

System Practices

Smith instructional computer systems are meant to be used for any and every educational purpose. There are no fixed limits on or charges for login time, use of the CPU (processor), or file storage space. Since some resources can be used up—printer paper, or access to terminals—we ask that you use these only as necessary. Help keep the systems unregulated—don’t waste or monopolize resources. When file storage space runs out, pitch in to help make space available. Similarly, use your good judgement to minimize use of CPU-intensive programs when the system is heavily loaded. Use the system as much as you need to, but let other people work too. Beowulf, grendel, and the other Computer Science machines are run as an open system. The UNIX philosophy is to leave as much of the computer open for inspection as possible. One of the best ways to learn what’s available is to explore. Secrecy is not the standard on UNIX. However, while system files are public, other people’s files are not. Respect the privacy of other users. This account is for your use only; it may not be given away or shared.

Getting Started

After you first log on, you might want to change your given, random password; and you might want to update your ‘‘finger’’information. Be forewarned that passwd will check for and reject easily cracked passwords. To change your password, type: passwd To change your ‘‘finger’’ information, type: chfn To logout, type: logout A few other Unix commands: To check your finger information, type: finger 112b-ak To see manual page for command, type: man command To list commands related to keyword, type: man -k keyword To list a text file, type: more To read mail, type: pine To read news, type: tin Other commands (use man to learn more): ls, chmod, finger, w, mkdir, cd, mv, emacs, vi ........................................................................................................................................................................ Return this portion to your instructor. Account name: 112b-ak

In accepting this account, I agree to follow the system practices outlined above.

Name (please print):

Campus Box: Phone Number:

UNIX Course Account Form for 112b

Department of Computer Science, Smith College, Northampton, Massachusetts 01063

Here is your Computer Science UNIX account. Your account name is: 112b-al

Your password is: xfe15=ER

Read the material on ‘‘System Practices,’’ sign the agreement below, tear it off along the dotted line, and return it to your instructor.

System Practices

Smith instructional computer systems are meant to be used for any and every educational purpose. There are no fixed limits on or charges for login time, use of the CPU (processor), or file storage space. Since some resources can be used up—printer paper, or access to terminals—we ask that you use these only as necessary. Help keep the systems unregulated—don’t waste or monopolize resources. When file storage space runs out, pitch in to help make space available. Similarly, use your good judgement to minimize use of CPU-intensive programs when the system is heavily loaded. Use the system as much as you need to, but let other people work too. Beowulf, grendel, and the other Computer Science machines are run as an open system. The UNIX philosophy is to leave as much of the computer open for inspection as possible. One of the best ways to learn what’s available is to explore. Secrecy is not the standard on UNIX. However, while system files are public, other people’s files are not. Respect the privacy of other users. This account is for your use only; it may not be given away or shared.

Getting Started

After you first log on, you might want to change your given, random password; and you might want to update your ‘‘finger’’information. Be forewarned that passwd will check for and reject easily cracked passwords. To change your password, type: passwd To change your ‘‘finger’’ information, type: chfn To logout, type: logout A few other Unix commands: To check your finger information, type: finger 112b-al To see manual page for command, type: man command To list commands related to keyword, type: man -k keyword To list a text file, type: more To read mail, type: pine To read news, type: tin Other commands (use man to learn more): ls, chmod, finger, w, mkdir, cd, mv, emacs, vi ........................................................................................................................................................................ Return this portion to your instructor. Account name: 112b-al

In accepting this account, I agree to follow the system practices outlined above.

Name (please print):

Campus Box: Phone Number:

UNIX Course Account Form for 112b

Department of Computer Science, Smith College, Northampton, Massachusetts 01063

Here is your Computer Science UNIX account. Your account name is: 112b-am

Your password is: TZtn6{p3

Read the material on ‘‘System Practices,’’ sign the agreement below, tear it off along the dotted line, and return it to your instructor.

System Practices

Smith instructional computer systems are meant to be used for any and every educational purpose. There are no fixed limits on or charges for login time, use of the CPU (processor), or file storage space. Since some resources can be used up—printer paper, or access to terminals—we ask that you use these only as necessary. Help keep the systems unregulated—don’t waste or monopolize resources. When file storage space runs out, pitch in to help make space available. Similarly, use your good judgement to minimize use of CPU-intensive programs when the system is heavily loaded. Use the system as much as you need to, but let other people work too. Beowulf, grendel, and the other Computer Science machines are run as an open system. The UNIX philosophy is to leave as much of the computer open for inspection as possible. One of the best ways to learn what’s available is to explore. Secrecy is not the standard on UNIX. However, while system files are public, other people’s files are not. Respect the privacy of other users. This account is for your use only; it may not be given away or shared.

Getting Started

After you first log on, you might want to change your given, random password; and you might want to update your ‘‘finger’’information. Be forewarned that passwd will check for and reject easily cracked passwords. To change your password, type: passwd To change your ‘‘finger’’ information, type: chfn To logout, type: logout A few other Unix commands: To check your finger information, type: finger 112b-am To see manual page for command, type: man command To list commands related to keyword, type: man -k keyword To list a text file, type: more To read mail, type: pine To read news, type: tin Other commands (use man to learn more): ls, chmod, finger, w, mkdir, cd, mv, emacs, vi ........................................................................................................................................................................ Return this portion to your instructor. Account name: 112b-am

In accepting this account, I agree to follow the system practices outlined above.

Name (please print):

Campus Box: Phone Number:

UNIX Course Account Form for 112b

Department of Computer Science, Smith College, Northampton, Massachusetts 01063

Here is your Computer Science UNIX account. Your account name is: 112b-an

Your password is: o6IE4uu@

Read the material on ‘‘System Practices,’’ sign the agreement below, tear it off along the dotted line, and return it to your instructor.

System Practices

Smith instructional computer systems are meant to be used for any and every educational purpose. There are no fixed limits on or charges for login time, use of the CPU (processor), or file storage space. Since some resources can be used up—printer paper, or access to terminals—we ask that you use these only as necessary. Help keep the systems unregulated—don’t waste or monopolize resources. When file storage space runs out, pitch in to help make space available. Similarly, use your good judgement to minimize use of CPU-intensive programs when the system is heavily loaded. Use the system as much as you need to, but let other people work too. Beowulf, grendel, and the other Computer Science machines are run as an open system. The UNIX philosophy is to leave as much of the computer open for inspection as possible. One of the best ways to learn what’s available is to explore. Secrecy is not the standard on UNIX. However, while system files are public, other people’s files are not. Respect the privacy of other users. This account is for your use only; it may not be given away or shared.

Getting Started

After you first log on, you might want to change your given, random password; and you might want to update your ‘‘finger’’information. Be forewarned that passwd will check for and reject easily cracked passwords. To change your password, type: passwd To change your ‘‘finger’’ information, type: chfn To logout, type: logout A few other Unix commands: To check your finger information, type: finger 112b-an To see manual page for command, type: man command To list commands related to keyword, type: man -k keyword To list a text file, type: more To read mail, type: pine To read news, type: tin Other commands (use man to learn more): ls, chmod, finger, w, mkdir, cd, mv, emacs, vi ........................................................................................................................................................................ Return this portion to your instructor. Account name: 112b-an

In accepting this account, I agree to follow the system practices outlined above.

Name (please print):

Campus Box: Phone Number:

UNIX Course Account Form for 112b

Department of Computer Science, Smith College, Northampton, Massachusetts 01063

Here is your Computer Science UNIX account. Your account name is: 112b-ao

Your password is: jh91FFg@

Read the material on ‘‘System Practices,’’ sign the agreement below, tear it off along the dotted line, and return it to your instructor.

System Practices

Smith instructional computer systems are meant to be used for any and every educational purpose. There are no fixed limits on or charges for login time, use of the CPU (processor), or file storage space. Since some resources can be used up—printer paper, or access to terminals—we ask that you use these only as necessary. Help keep the systems unregulated—don’t waste or monopolize resources. When file storage space runs out, pitch in to help make space available. Similarly, use your good judgement to minimize use of CPU-intensive programs when the system is heavily loaded. Use the system as much as you need to, but let other people work too. Beowulf, grendel, and the other Computer Science machines are run as an open system. The UNIX philosophy is to leave as much of the computer open for inspection as possible. One of the best ways to learn what’s available is to explore. Secrecy is not the standard on UNIX. However, while system files are public, other people’s files are not. Respect the privacy of other users. This account is for your use only; it may not be given away or shared.

Getting Started

After you first log on, you might want to change your given, random password; and you might want to update your ‘‘finger’’information. Be forewarned that passwd will check for and reject easily cracked passwords. To change your password, type: passwd To change your ‘‘finger’’ information, type: chfn To logout, type: logout A few other Unix commands: To check your finger information, type: finger 112b-ao To see manual page for command, type: man command To list commands related to keyword, type: man -k keyword To list a text file, type: more To read mail, type: pine To read news, type: tin Other commands (use man to learn more): ls, chmod, finger, w, mkdir, cd, mv, emacs, vi ........................................................................................................................................................................ Return this portion to your instructor. Account name: 112b-ao

In accepting this account, I agree to follow the system practices outlined above.

Name (please print):

Campus Box: Phone Number:

UNIX Course Account Form for 112b

Department of Computer Science, Smith College, Northampton, Massachusetts 01063

Here is your Computer Science UNIX account. Your account name is: 112b-ap

Your password is: Suj81<tA

Read the material on ‘‘System Practices,’’ sign the agreement below, tear it off along the dotted line, and return it to your instructor.

System Practices

Smith instructional computer systems are meant to be used for any and every educational purpose. There are no fixed limits on or charges for login time, use of the CPU (processor), or file storage space. Since some resources can be used up—printer paper, or access to terminals—we ask that you use these only as necessary. Help keep the systems unregulated—don’t waste or monopolize resources. When file storage space runs out, pitch in to help make space available. Similarly, use your good judgement to minimize use of CPU-intensive programs when the system is heavily loaded. Use the system as much as you need to, but let other people work too. Beowulf, grendel, and the other Computer Science machines are run as an open system. The UNIX philosophy is to leave as much of the computer open for inspection as possible. One of the best ways to learn what’s available is to explore. Secrecy is not the standard on UNIX. However, while system files are public, other people’s files are not. Respect the privacy of other users. This account is for your use only; it may not be given away or shared.

Getting Started

After you first log on, you might want to change your given, random password; and you might want to update your ‘‘finger’’information. Be forewarned that passwd will check for and reject easily cracked passwords. To change your password, type: passwd To change your ‘‘finger’’ information, type: chfn To logout, type: logout A few other Unix commands: To check your finger information, type: finger 112b-ap To see manual page for command, type: man command To list commands related to keyword, type: man -k keyword To list a text file, type: more To read mail, type: pine To read news, type: tin Other commands (use man to learn more): ls, chmod, finger, w, mkdir, cd, mv, emacs, vi ........................................................................................................................................................................ Return this portion to your instructor. Account name: 112b-ap

In accepting this account, I agree to follow the system practices outlined above.

Name (please print):

Campus Box: Phone Number:

UNIX Course Account Form for 112b

Department of Computer Science, Smith College, Northampton, Massachusetts 01063

Here is your Computer Science UNIX account. Your account name is: 112b-aq

Your password is: mXQ3i~1n

Read the material on ‘‘System Practices,’’ sign the agreement below, tear it off along the dotted line, and return it to your instructor.

System Practices

Smith instructional computer systems are meant to be used for any and every educational purpose. There are no fixed limits on or charges for login time, use of the CPU (processor), or file storage space. Since some resources can be used up—printer paper, or access to terminals—we ask that you use these only as necessary. Help keep the systems unregulated—don’t waste or monopolize resources. When file storage space runs out, pitch in to help make space available. Similarly, use your good judgement to minimize use of CPU-intensive programs when the system is heavily loaded. Use the system as much as you need to, but let other people work too. Beowulf, grendel, and the other Computer Science machines are run as an open system. The UNIX philosophy is to leave as much of the computer open for inspection as possible. One of the best ways to learn what’s available is to explore. Secrecy is not the standard on UNIX. However, while system files are public, other people’s files are not. Respect the privacy of other users. This account is for your use only; it may not be given away or shared.

Getting Started

After you first log on, you might want to change your given, random password; and you might want to update your ‘‘finger’’information. Be forewarned that passwd will check for and reject easily cracked passwords. To change your password, type: passwd To change your ‘‘finger’’ information, type: chfn To logout, type: logout A few other Unix commands: To check your finger information, type: finger 112b-aq To see manual page for command, type: man command To list commands related to keyword, type: man -k keyword To list a text file, type: more To read mail, type: pine To read news, type: tin Other commands (use man to learn more): ls, chmod, finger, w, mkdir, cd, mv, emacs, vi ........................................................................................................................................................................ Return this portion to your instructor. Account name: 112b-aq

In accepting this account, I agree to follow the system practices outlined above.

Name (please print):

Campus Box: Phone Number:

UNIX Course Account Form for 112b

Department of Computer Science, Smith College, Northampton, Massachusetts 01063

Here is your Computer Science UNIX account. Your account name is: 112b-ar

Your password is: 62xWF:jn

Read the material on ‘‘System Practices,’’ sign the agreement below, tear it off along the dotted line, and return it to your instructor.

System Practices

Smith instructional computer systems are meant to be used for any and every educational purpose. There are no fixed limits on or charges for login time, use of the CPU (processor), or file storage space. Since some resources can be used up—printer paper, or access to terminals—we ask that you use these only as necessary. Help keep the systems unregulated—don’t waste or monopolize resources. When file storage space runs out, pitch in to help make space available. Similarly, use your good judgement to minimize use of CPU-intensive programs when the system is heavily loaded. Use the system as much as you need to, but let other people work too. Beowulf, grendel, and the other Computer Science machines are run as an open system. The UNIX philosophy is to leave as much of the computer open for inspection as possible. One of the best ways to learn what’s available is to explore. Secrecy is not the standard on UNIX. However, while system files are public, other people’s files are not. Respect the privacy of other users. This account is for your use only; it may not be given away or shared.

Getting Started

After you first log on, you might want to change your given, random password; and you might want to update your ‘‘finger’’information. Be forewarned that passwd will check for and reject easily cracked passwords. To change your password, type: passwd To change your ‘‘finger’’ information, type: chfn To logout, type: logout A few other Unix commands: To check your finger information, type: finger 112b-ar To see manual page for command, type: man command To list commands related to keyword, type: man -k keyword To list a text file, type: more To read mail, type: pine To read news, type: tin Other commands (use man to learn more): ls, chmod, finger, w, mkdir, cd, mv, emacs, vi ........................................................................................................................................................................ Return this portion to your instructor. Account name: 112b-ar

In accepting this account, I agree to follow the system practices outlined above.

Name (please print):

Campus Box: Phone Number:

UNIX Course Account Form for 112b

Department of Computer Science, Smith College, Northampton, Massachusetts 01063

Here is your Computer Science UNIX account. Your account name is: 112b-as

Your password is: e+Hr2Bv5

Read the material on ‘‘System Practices,’’ sign the agreement below, tear it off along the dotted line, and return it to your instructor.

System Practices

Smith instructional computer systems are meant to be used for any and every educational purpose. There are no fixed limits on or charges for login time, use of the CPU (processor), or file storage space. Since some resources can be used up—printer paper, or access to terminals—we ask that you use these only as necessary. Help keep the systems unregulated—don’t waste or monopolize resources. When file storage space runs out, pitch in to help make space available. Similarly, use your good judgement to minimize use of CPU-intensive programs when the system is heavily loaded. Use the system as much as you need to, but let other people work too. Beowulf, grendel, and the other Computer Science machines are run as an open system. The UNIX philosophy is to leave as much of the computer open for inspection as possible. One of the best ways to learn what’s available is to explore. Secrecy is not the standard on UNIX. However, while system files are public, other people’s files are not. Respect the privacy of other users. This account is for your use only; it may not be given away or shared.

Getting Started

After you first log on, you might want to change your given, random password; and you might want to update your ‘‘finger’’information. Be forewarned that passwd will check for and reject easily cracked passwords. To change your password, type: passwd To change your ‘‘finger’’ information, type: chfn To logout, type: logout A few other Unix commands: To check your finger information, type: finger 112b-as To see manual page for command, type: man command To list commands related to keyword, type: man -k keyword To list a text file, type: more To read mail, type: pine To read news, type: tin Other commands (use man to learn more): ls, chmod, finger, w, mkdir, cd, mv, emacs, vi ........................................................................................................................................................................ Return this portion to your instructor. Account name: 112b-as

In accepting this account, I agree to follow the system practices outlined above.

Name (please print):

Campus Box: Phone Number:

UNIX Course Account Form for 112b

Department of Computer Science, Smith College, Northampton, Massachusetts 01063

Here is your Computer Science UNIX account. Your account name is: 112b-at

Your password is: vb1OD|m4

Read the material on ‘‘System Practices,’’ sign the agreement below, tear it off along the dotted line, and return it to your instructor.

System Practices

Smith instructional computer systems are meant to be used for any and every educational purpose. There are no fixed limits on or charges for login time, use of the CPU (processor), or file storage space. Since some resources can be used up—printer paper, or access to terminals—we ask that you use these only as necessary. Help keep the systems unregulated—don’t waste or monopolize resources. When file storage space runs out, pitch in to help make space available. Similarly, use your good judgement to minimize use of CPU-intensive programs when the system is heavily loaded. Use the system as much as you need to, but let other people work too. Beowulf, grendel, and the other Computer Science machines are run as an open system. The UNIX philosophy is to leave as much of the computer open for inspection as possible. One of the best ways to learn what’s available is to explore. Secrecy is not the standard on UNIX. However, while system files are public, other people’s files are not. Respect the privacy of other users. This account is for your use only; it may not be given away or shared.

Getting Started

After you first log on, you might want to change your given, random password; and you might want to update your ‘‘finger’’information. Be forewarned that passwd will check for and reject easily cracked passwords. To change your password, type: passwd To change your ‘‘finger’’ information, type: chfn To logout, type: logout A few other Unix commands: To check your finger information, type: finger 112b-at To see manual page for command, type: man command To list commands related to keyword, type: man -k keyword To list a text file, type: more To read mail, type: pine To read news, type: tin Other commands (use man to learn more): ls, chmod, finger, w, mkdir, cd, mv, emacs, vi ........................................................................................................................................................................ Return this portion to your instructor. Account name: 112b-at

In accepting this account, I agree to follow the system practices outlined above.

Name (please print):

Campus Box: Phone Number:

UNIX Course Account Form for 112b

Department of Computer Science, Smith College, Northampton, Massachusetts 01063

Here is your Computer Science UNIX account. Your account name is: 112b-au

Your password is: r8mlZA6&

Read the material on ‘‘System Practices,’’ sign the agreement below, tear it off along the dotted line, and return it to your instructor.

System Practices

Smith instructional computer systems are meant to be used for any and every educational purpose. There are no fixed limits on or charges for login time, use of the CPU (processor), or file storage space. Since some resources can be used up—printer paper, or access to terminals—we ask that you use these only as necessary. Help keep the systems unregulated—don’t waste or monopolize resources. When file storage space runs out, pitch in to help make space available. Similarly, use your good judgement to minimize use of CPU-intensive programs when the system is heavily loaded. Use the system as much as you need to, but let other people work too. Beowulf, grendel, and the other Computer Science machines are run as an open system. The UNIX philosophy is to leave as much of the computer open for inspection as possible. One of the best ways to learn what’s available is to explore. Secrecy is not the standard on UNIX. However, while system files are public, other people’s files are not. Respect the privacy of other users. This account is for your use only; it may not be given away or shared.

Getting Started

After you first log on, you might want to change your given, random password; and you might want to update your ‘‘finger’’information. Be forewarned that passwd will check for and reject easily cracked passwords. To change your password, type: passwd To change your ‘‘finger’’ information, type: chfn To logout, type: logout A few other Unix commands: To check your finger information, type: finger 112b-au To see manual page for command, type: man command To list commands related to keyword, type: man -k keyword To list a text file, type: more To read mail, type: pine To read news, type: tin Other commands (use man to learn more): ls, chmod, finger, w, mkdir, cd, mv, emacs, vi ........................................................................................................................................................................ Return this portion to your instructor. Account name: 112b-au

In accepting this account, I agree to follow the system practices outlined above.

Name (please print):

Campus Box: Phone Number:

UNIX Course Account Form for 112b

Department of Computer Science, Smith College, Northampton, Massachusetts 01063

Here is your Computer Science UNIX account. Your account name is: 112b-av

Your password is: icF^32cD

Read the material on ‘‘System Practices,’’ sign the agreement below, tear it off along the dotted line, and return it to your instructor.

System Practices

Smith instructional computer systems are meant to be used for any and every educational purpose. There are no fixed limits on or charges for login time, use of the CPU (processor), or file storage space. Since some resources can be used up—printer paper, or access to terminals—we ask that you use these only as necessary. Help keep the systems unregulated—don’t waste or monopolize resources. When file storage space runs out, pitch in to help make space available. Similarly, use your good judgement to minimize use of CPU-intensive programs when the system is heavily loaded. Use the system as much as you need to, but let other people work too. Beowulf, grendel, and the other Computer Science machines are run as an open system. The UNIX philosophy is to leave as much of the computer open for inspection as possible. One of the best ways to learn what’s available is to explore. Secrecy is not the standard on UNIX. However, while system files are public, other people’s files are not. Respect the privacy of other users. This account is for your use only; it may not be given away or shared.

Getting Started

After you first log on, you might want to change your given, random password; and you might want to update your ‘‘finger’’information. Be forewarned that passwd will check for and reject easily cracked passwords. To change your password, type: passwd To change your ‘‘finger’’ information, type: chfn To logout, type: logout A few other Unix commands: To check your finger information, type: finger 112b-av To see manual page for command, type: man command To list commands related to keyword, type: man -k keyword To list a text file, type: more To read mail, type: pine To read news, type: tin Other commands (use man to learn more): ls, chmod, finger, w, mkdir, cd, mv, emacs, vi ........................................................................................................................................................................ Return this portion to your instructor. Account name: 112b-av

In accepting this account, I agree to follow the system practices outlined above.

Name (please print):

Campus Box: Phone Number:

UNIX Course Account Form for 112b

Department of Computer Science, Smith College, Northampton, Massachusetts 01063

Here is your Computer Science UNIX account. Your account name is: 112b-aw

Your password is: 0yPJy6t?

Read the material on ‘‘System Practices,’’ sign the agreement below, tear it off along the dotted line, and return it to your instructor.

System Practices

Smith instructional computer systems are meant to be used for any and every educational purpose. There are no fixed limits on or charges for login time, use of the CPU (processor), or file storage space. Since some resources can be used up—printer paper, or access to terminals—we ask that you use these only as necessary. Help keep the systems unregulated—don’t waste or monopolize resources. When file storage space runs out, pitch in to help make space available. Similarly, use your good judgement to minimize use of CPU-intensive programs when the system is heavily loaded. Use the system as much as you need to, but let other people work too. Beowulf, grendel, and the other Computer Science machines are run as an open system. The UNIX philosophy is to leave as much of the computer open for inspection as possible. One of the best ways to learn what’s available is to explore. Secrecy is not the standard on UNIX. However, while system files are public, other people’s files are not. Respect the privacy of other users. This account is for your use only; it may not be given away or shared.

Getting Started

After you first log on, you might want to change your given, random password; and you might want to update your ‘‘finger’’information. Be forewarned that passwd will check for and reject easily cracked passwords. To change your password, type: passwd To change your ‘‘finger’’ information, type: chfn To logout, type: logout A few other Unix commands: To check your finger information, type: finger 112b-aw To see manual page for command, type: man command To list commands related to keyword, type: man -k keyword To list a text file, type: more To read mail, type: pine To read news, type: tin Other commands (use man to learn more): ls, chmod, finger, w, mkdir, cd, mv, emacs, vi ........................................................................................................................................................................ Return this portion to your instructor. Account name: 112b-aw

In accepting this account, I agree to follow the system practices outlined above.

Name (please print):

Campus Box: Phone Number:

UNIX Course Account Form for 112b

Department of Computer Science, Smith College, Northampton, Massachusetts 01063

Here is your Computer Science UNIX account. Your account name is: 112b-ax

Your password is: cDD"xc65

Read the material on ‘‘System Practices,’’ sign the agreement below, tear it off along the dotted line, and return it to your instructor.

System Practices

Smith instructional computer systems are meant to be used for any and every educational purpose. There are no fixed limits on or charges for login time, use of the CPU (processor), or file storage space. Since some resources can be used up—printer paper, or access to terminals—we ask that you use these only as necessary. Help keep the systems unregulated—don’t waste or monopolize resources. When file storage space runs out, pitch in to help make space available. Similarly, use your good judgement to minimize use of CPU-intensive programs when the system is heavily loaded. Use the system as much as you need to, but let other people work too. Beowulf, grendel, and the other Computer Science machines are run as an open system. The UNIX philosophy is to leave as much of the computer open for inspection as possible. One of the best ways to learn what’s available is to explore. Secrecy is not the standard on UNIX. However, while system files are public, other people’s files are not. Respect the privacy of other users. This account is for your use only; it may not be given away or shared.

Getting Started

After you first log on, you might want to change your given, random password; and you might want to update your ‘‘finger’’information. Be forewarned that passwd will check for and reject easily cracked passwords. To change your password, type: passwd To change your ‘‘finger’’ information, type: chfn To logout, type: logout A few other Unix commands: To check your finger information, type: finger 112b-ax To see manual page for command, type: man command To list commands related to keyword, type: man -k keyword To list a text file, type: more To read mail, type: pine To read news, type: tin Other commands (use man to learn more): ls, chmod, finger, w, mkdir, cd, mv, emacs, vi ........................................................................................................................................................................ Return this portion to your instructor. Account name: 112b-ax

In accepting this account, I agree to follow the system practices outlined above.

Name (please print):

Campus Box: Phone Number:

UNIX Course Account Form for 112b

Department of Computer Science, Smith College, Northampton, Massachusetts 01063

Here is your Computer Science UNIX account. Your account name is: 112b-ay

Your password is: I’61ggfZ

Read the material on ‘‘System Practices,’’ sign the agreement below, tear it off along the dotted line, and return it to your instructor.

System Practices

Smith instructional computer systems are meant to be used for any and every educational purpose. There are no fixed limits on or charges for login time, use of the CPU (processor), or file storage space. Since some resources can be used up—printer paper, or access to terminals—we ask that you use these only as necessary. Help keep the systems unregulated—don’t waste or monopolize resources. When file storage space runs out, pitch in to help make space available. Similarly, use your good judgement to minimize use of CPU-intensive programs when the system is heavily loaded. Use the system as much as you need to, but let other people work too. Beowulf, grendel, and the other Computer Science machines are run as an open system. The UNIX philosophy is to leave as much of the computer open for inspection as possible. One of the best ways to learn what’s available is to explore. Secrecy is not the standard on UNIX. However, while system files are public, other people’s files are not. Respect the privacy of other users. This account is for your use only; it may not be given away or shared.

Getting Started

After you first log on, you might want to change your given, random password; and you might want to update your ‘‘finger’’information. Be forewarned that passwd will check for and reject easily cracked passwords. To change your password, type: passwd To change your ‘‘finger’’ information, type: chfn To logout, type: logout A few other Unix commands: To check your finger information, type: finger 112b-ay To see manual page for command, type: man command To list commands related to keyword, type: man -k keyword To list a text file, type: more To read mail, type: pine To read news, type: tin Other commands (use man to learn more): ls, chmod, finger, w, mkdir, cd, mv, emacs, vi ........................................................................................................................................................................ Return this portion to your instructor. Account name: 112b-ay

In accepting this account, I agree to follow the system practices outlined above.

Name (please print):

Campus Box: Phone Number:

UNIX Course Account Form for 112b

Department of Computer Science, Smith College, Northampton, Massachusetts 01063

Here is your Computer Science UNIX account. Your account name is: 112b-az

Your password is: Px0Jv[p6

Read the material on ‘‘System Practices,’’ sign the agreement below, tear it off along the dotted line, and return it to your instructor.

System Practices

Smith instructional computer systems are meant to be used for any and every educational purpose. There are no fixed limits on or charges for login time, use of the CPU (processor), or file storage space. Since some resources can be used up—printer paper, or access to terminals—we ask that you use these only as necessary. Help keep the systems unregulated—don’t waste or monopolize resources. When file storage space runs out, pitch in to help make space available. Similarly, use your good judgement to minimize use of CPU-intensive programs when the system is heavily loaded. Use the system as much as you need to, but let other people work too. Beowulf, grendel, and the other Computer Science machines are run as an open system. The UNIX philosophy is to leave as much of the computer open for inspection as possible. One of the best ways to learn what’s available is to explore. Secrecy is not the standard on UNIX. However, while system files are public, other people’s files are not. Respect the privacy of other users. This account is for your use only; it may not be given away or shared.

Getting Started

After you first log on, you might want to change your given, random password; and you might want to update your ‘‘finger’’information. Be forewarned that passwd will check for and reject easily cracked passwords. To change your password, type: passwd To change your ‘‘finger’’ information, type: chfn To logout, type: logout A few other Unix commands: To check your finger information, type: finger 112b-az To see manual page for command, type: man command To list commands related to keyword, type: man -k keyword To list a text file, type: more To read mail, type: pine To read news, type: tin Other commands (use man to learn more): ls, chmod, finger, w, mkdir, cd, mv, emacs, vi ........................................................................................................................................................................ Return this portion to your instructor. Account name: 112b-az

In accepting this account, I agree to follow the system practices outlined above.

Name (please print):

Campus Box: Phone Number: